Author's Response To Reviewer Comments



mixing the wrong indices (handling 48 tubes is much more error prone than handling just three 8-well stripes).

In addition to this rather practical lab-work related reason we highlighted (Line: 602 to 605) that it is still very unlikely that the repeated use of tags for multiple samples causes accepted false positives in the end, as the final acceptance is not based on single occurrence but on repeated occurrence in independent replicates. We fully agree that the use of non-matching tags (e.g. A/B) would increase the number of samples that could be analysed in one sequencing run. But at the same time it would make it much harder to identify contaminations or tag jumps as we discuss in line 575 to 599. Contaminations of a PCR with another differently labelled PCR product would increase the number of chimeras in your PCR which would remain undetected if you would also use non-matching tag combination. The same holds true for tag-jumps, which are an issue in Illumina sequencing (see Schnell et al. 2015) and where we could demonstrate that our PCR libraries reduce the read-losses compared to adapter-ligation techniques (lines 585-594).Particular for the last reasons we favoured to use only twin-tag combinations.

We also thank the reviewer for the interesting paper, which also used quadruple-indexed libraries. We do however not see the application of RAD sequencing to identify invertebrate-derived DNA of unknown origin. Generally RAD sequencing requires high molecular weight genomic DNA. Our samples have a mixed pools of genomic and mitochondrial DNA from different organisms and our target DNA is often highly degraded, of poor quality and of low quantities. In addition we have the presence of high amounts of leech DNA. Therefore we currently do not see an application of this sequencing method.

"Also the read losses due to trimming and quality filtering were significantly lower for the 16S sequencing runs (1.3% and 5.3% in average, Supplemental Table 3) compared to the sequencing runs for the longer fragments of 12S and CytB (65.3% and 44.3% in average, Supplemental Table 3)."

The Usearch read overlapping pipeline is sensitive to number of mismatches in alignment. The Read2 in MiSeq 600 cycles run is particularly notoriously for being low quality towards the end of the run. Try trimming both R1 and R2 to 250 bp (length trimming) and redo the overlap and read loss calculation.

Thank you, for this valuable advice. We tested it for one of our 12S runs and compared results. As you suggested we trimmed the reads to 250 base pairs adjusted the -fastq_minovlen parameter for usearch from 50bp to just 25bp as we would expect to have a smaller overlap of the trimmed reads. In fact we obtained more read after merging (13,129,505 vs. 13,388,933). However, most of those reads were lost again after filtering so that our original settings produced in fact the most reads I the end (4,694,624 vs. 4,227,346). Thus we think it is reasonable to stick to the current settings in the pipeline. Results original pipeline: raw reads: 13,766,169 merging: 13,129,505 clipping: 6,498,738 filtering: 4,694,624 Trimmed reads (trimm 250bp, overlap 25bp): raw reads: 13,766,169 merging: 13,388,933 clipping: 6,684,766

"All three markers were amplified simultaneously for each batch of samples in a single PCR plate".

In different individual well?

filterina: 4,227,346

Sorry for the misunderstanding, we did not do multiplex-PCR and amplified in individual wells. We added this to the sentence in lines 324-325:

"... All three markers were amplified simultaneously in individual wells for each batch of samples in a single PCR plate. ..."

Because of different amplicon lengths and therefore different binding affinities to the flow cell Also due to clustering efficiency . smaller fragment = easier to amplify # We agree, also due to DNA degradation we had higher amplification success for the shortest fragment (see lines 562 – 566). As we say in lines 337-340 "...Because of different amplicon lengths and therefore different binding affinities to the flow cell, 12S and CytB products were combined in a single library, whereas positive 16S products were always combined in a separate library. ..." and these libraries were sequenced independently. To make this clearer we added a second sentence (line 340): "... 12S/CytB libraries were sequenced independently from 16S libraries...."

Clo <u>s</u> e
